Mon Cct 19 16:41:53 GMr 1998
Year: 98 Doy: 292
Qobserver: yasukawa
WEATHER COMMVENT: Mon Cct 19 16:57:34 GMIr 1998
Col d, clear, no w nd.
Frost on ground.
COVMENT: Mon Cct 19 16:58:22 GV 1998
Kim and David here to install MIV, spar systens down.
Gbserving with LOAL and PSPT only.
WEATHER COMMVENT: Mon Cct 19 20:54:43 GMr 1998
I n orographic clouds. Stopped PSPT.
COMMENT: Mon Cct 19 20:58:33 GMI 1998
CGoi ng down to GONG
COMMENT: Tue Cct 20 01:13:24 GVIT 1998
Found | ost start pul ses, software was conpiled for Boul der and not
for here. David fixidit.
CH P filter fixed.
Excess circuitry and wiring removed from MII1 barrel
Chip filter polarizers replaced and tenperature controller fixed.
COMMENT: Tue Cct 20 01:16: 09 GVI 1998
Tapes:

LOAL:  LO0549
Tue Cct 20 01:17: 09 GMI 1998
M1 T



